\newcommand{\noopsort}[1]{} \newcommand{\printfirst}[2]{#1}
  \newcommand{\singleletter}[1]{#1} \newcommand{\switchargs}[2]{#2#1}
\begin{thebibliography}{10}

\bibitem{altintas2006kepler}
Ilkay Altintas, Oscar Barney, and Efrat Jaeger-Frank.
\newblock Provenance collection support in the kepler scientific workflow
  system.
\newblock In {\em Provenance and Annotation of Data: International Provenance
  and Annotation Workshop, IPAW 2006}, volume 4145/2006 of {\em Lecture Notes
  in Computer Science}, pages {118--132}, Chicago, IL, USA, May 2006. Springer.

\bibitem{anderson2004boinc}
David~P. Anderson.
\newblock {BOINC}: A system for public-resource computing and storage.
\newblock In {\em Proceedings of the 5th IEEE/ACM International Workshop on
  Grid Computing}, pages {4--10}. IEEE Computer Society, 2004.

\bibitem{anderson2002seti}
David~P. Anderson, Jeff Cobb, Eric Korpela, Matt Lebofsky, and Dan Werthimer.
\newblock {SETI@home}: An experiment in public-resource computing.
\newblock {\em Communications of the ACM}, 45(11):{56--61}, November 2002.

\bibitem{awerbuch1984reliable}
Baruch Awerbuch and Shimon Even.
\newblock Efficient and reliable broadcast is achievable in an eventually
  connected network.
\newblock In {\em Proceedings of the Third Annual ACM Symposium on Principles
  of Distributed Computing}, pages {278--281}. ACM, 1984.

\bibitem{bellman1961processes}
Richard~E. Bellman.
\newblock {\em Adaptive Control Processes: A Guided Tour}.
\newblock Princeton University Press, Princeton, NJ, USA, 1961.

\bibitem{rfc1738}
T.~Berners-Lee, L.~Masinter, and M.~McCahill.
\newblock {RFC 1738}: {Uniform Resource Locators (URL)}, December 1994.

\bibitem{birrell1982grapevine}
Andrew~D. Birrell, Roy Levin, Roger~M. Needham, and Michael~D. Schroeder.
\newblock Grapevine: An exercise in distributed computing.
\newblock {\em Communications of the ACM}, 25(4):{260--274}, April 1982.

\bibitem{braun2006provsec}
Uri Braun and Avi Shinnar.
\newblock A security model for provenance.
\newblock Technical Report TR-04-06, Computer Science Group, Harvard
  University, 2006.

\bibitem{buneman2001whywhere}
Peter Buneman, Sanjeev Khanna, and Wang-Chiew Tan.
\newblock Why and where: A characterization of data provenance.
\newblock In {\em ICDT '01: Proceedings of the 8th International Conference on
  Database Theory}, volume 1973 of {\em Lecture Notes in Computer Science},
  pages {316--330}, London, UK, 2001. Springer.

\bibitem{cerf1974tcp}
Vinton~G. Cerf and Robert~E. Kahn.
\newblock A protocol for packet network intercommunication.
\newblock {\em IEEE Transactions on Communications}, 22(5):{637--648}, May
  1974.

\bibitem{dnssdDraft}
Stuart Cheshire and Marc Krochmal.
\newblock {DNS}-based service discovery, March 2010.

\bibitem{dacruz2008provenanceservices}
S\'{e}rgio Manuel~Serra da~Cruz, Patr\'{\i}cia~M. Barros, Paulo~M. Bisch, Maria
  Luiza~Machado Campos, and Marta Mattoso.
\newblock Provenance services for distributed workflows.
\newblock In {\em Proceedings of the 2008 Eighth IEEE International Symposium
  on Cluster Computing and the Grid}, pages {526--533}. IEEE Computer Society,
  2008.

\bibitem{dean2004mapreduce}
Jeffrey Dean and Sanjay Ghemawat.
\newblock {MapReduce}: Simplified data processing on large clusters.
\newblock In {\em Proceedings of the 6th Symposium on Operating System Design
  and Implementation (OSDI 2004)}. USENIX Association, 2004.

\bibitem{demers1987epidemic}
Alan Demers, Dan Greene, Carl Hauser, Wes Irish, John Larson, Scott Shenker,
  Howard Sturgis, Dan Swinehart, and Doug Terry.
\newblock Epidemic algorithms for replicated database maintenance.
\newblock In {\em Proceedings of the Sixth Annual ACM Symposium on Principles
  of Distributed Computing}, pages {1--12}. ACM, 1987.

\bibitem{doorenbos1995thesis}
Robert~B. Doorenbos.
\newblock {\em Production Matching for Large Learning Systems}.
\newblock PhD thesis, Carnegie Mellon University, Pittsburgh, PA, USA, January
  1995.

\bibitem{rfc5789}
L.~Dusseault.
\newblock {RFC 5789}: {Patch Method for HTTP}, March 2010.

\bibitem{rfc2616}
R.~Fielding, J.~Gettys, J.~Mogul, H.~Frystyk, L.~Masinter, P.~Leach, and
  T.~Berners-Lee.
\newblock {RFC 2616}: Hypertext transfer protocol -- {HTTP/1.1}, June 1999.

\bibitem{fielding2000thesis}
Roy~Thomas Fielding.
\newblock {\em Architectural Styles and the Design of Network-based Software
  Architectures}.
\newblock PhD thesis, University of California, Irvine, Irvine, CA, USA, 2000.

\bibitem{forgy1979rete}
Charles~Lanny Forgy.
\newblock {\em On the Efficient Implementation of Production Systems}.
\newblock PhD thesis, Carnegie Mellon University, Pittsburgh, PA, USA, February
  1979.

\bibitem{friedman1976lazy}
Daniel~P. Friedman and David~S. Wise.
\newblock Cons should not evaluate its arguments.
\newblock In S.~Michaelson and R.~Milner, editors, {\em Automata, Languages and
  Programming}, pages {257--284}, Edinburgh, UK, 1976. Edinburgh University
  Press.

\bibitem{hasan2009picasso}
Ragib Hasan, Radu Sion, and Marianne Winslett.
\newblock The case of the fake {Picasso}: Preventing history forgery with
  secure provenance.
\newblock In {\em Proceedings of the 7th USENIX Conference on File and Storage
  Technologies (FAST 2009)}. USENIX Association, 2009.

\bibitem{khare2004arrested}
Rohit Khare and Richard~N. Taylor.
\newblock Extending the representational state transfer {(REST)} architectural
  style for distributed systems.
\newblock In {\em Proceedings of the 26th International Conference on Software
  Engineering}, pages {428--437}. IEEE Computer Society, 2004.

\bibitem{rfc4122}
P.~Leach, M.~Mealling, and R.~Salz.
\newblock {RFC 4122}: A {Universally Unique IDentifier (UUID) URN} namespace,
  July 2005.

\bibitem{rfc1034}
P.~Mockapetris.
\newblock {RFC 1034}: Domain names - concepts and facilities, November 1987.

\bibitem{rfc1035}
P.~Mockapetris.
\newblock {RFC 1035}: Domain names - implementation and specification, November
  1987.

\bibitem{moreau2008provenance}
Luc Moreau, Paul Groth, Simon Miles, Javier Vazquez-Salceda, John Ibbotson,
  Sheng Jiang, Steve Munroe, Omer Rana, Andreas Schreiber, Victor Tan, and
  Laszlo Varga.
\newblock The provenance of electronic data.
\newblock {\em Communications of the ACM}, 51(4):{52--58}, March 2008.

\bibitem{munroe2006prime}
Steve Munroe, Simon Miles, Luc Moreau, and Javier V\'{a}squez-Salceda.
\newblock {PrIMe:} a software engineering methodology for developing
  provenance-aware applications.
\newblock In {\em {SEM} '06: Proceedings of the 6th International Workshop on
  Software Engineering and Middleware}, pages {39--46}, Portland, OR, USA,
  September 2006. ACM.

\bibitem{narayanan2009social}
Arvind Narayanan and Vitaly Shmatikov.
\newblock De-anonymizing social networks.
\newblock In {\em 30th IEEE Symposium on Security and Privacy}, pages
  {173--187}, Oakland, CA, USA, 2009. IEEE Computer Society.

\bibitem{netzer1996mpirace}
Robert H.~B. Netzer, Timothy~W. Brennan, and Suresh~K. Damodaran-Kamal.
\newblock Debugging race conditions in message-passing programs.
\newblock In {\em Proceedings of the SIGMETRICS Symposium on Parallel and
  Distributed Tools}, pages {31--40}, Philadelphia, PA, USA, 1996. ACM.

\bibitem{netzer1992race}
Robert H.~B. Netzer and Barton~P. Miller.
\newblock What are race conditions?: Some issues and formalizations.
\newblock {\em ACM Letters on Programming Languages and Systems},
  1(1):{74--88}, March 1992.

\bibitem{parsons2004clustering}
Lance Parsons, Ehtesham Haque, and Huan Liu.
\newblock Subspace clustering for high dimensional data: A review.
\newblock {\em ACM SIGKDD Explorations Newsletter}, 6(1):{90--105}, 2004.

\bibitem{radul2009thesis}
Alexey Radul.
\newblock {\em Propagation Networks: A Flexible and Expressive Substrate for
  Computation}.
\newblock PhD thesis, Massachusetts Institute of Technology, 2009.

\bibitem{radul2009art}
Alexey Radul and Gerald~Jay Sussman.
\newblock The art of the propagator.
\newblock Technical Report MIT-CSAIL-TR-2009-002, MIT Computer Science and
  Artificial Intelligence Laboratory, January 2009.

\bibitem{sar2005lineagefs}
Can Sar and Pei Cao.
\newblock Lineage file system.
\newblock Department of Computer Science, Stanford University.
  http://crypto.stanford.edu/{\textasciitilde}cao/lineage.html, 2005.

\bibitem{schneier1999logs}
Bruce Schneier and John Kelsey.
\newblock Secure audit logs to support computer forensics.
\newblock {\em ACM Transactions on Information and System Security},
  2(2):159--176, 1999.

\bibitem{simmhan2005provenance}
Yogesh~L. Simmhan, Beth Plale, and Dennis Gannon.
\newblock A survey of data provenance in e-science.
\newblock {\em ACM SIGMOD Record}, 34(3):{31--36}, September 2005.

\bibitem{singhal1990updatetransport}
Mukesh Singhal.
\newblock Update transport: A new technique for update synchronization in
  replicated database systems.
\newblock {\em IEEE Transactions on Software Engineering}, 16(12):{1325--1336},
  December 1990.

\bibitem{szomszor2003ws}
Martin Szomszor and Luc Moreau.
\newblock Recording and reasoning over data provenance in web and grid
  services.
\newblock In {\em On the Move to Meaningful Internet Systems 2003: CoopIS, DOA,
  and ODBASE}, volume 2888/2003 of {\em Lecture Notes in Computer Science},
  pages {603--620}. Springer, 2003.

\bibitem{terry1995bayou}
Douglas~B. Terry, Marvin~M. Theimer, Karin Petersen, Alan~J. Demers, Mike~J.
  Spreitzer, and Carl~H. Hauser.
\newblock Managing update conflicts in bayou, a weakly connected replicated
  storage system.
\newblock In {\em Proceedings of the Fifteenth ACM Symposium on Operating
  Systems Principles}, pages {172--182}, Copper Mountain, CO, USA, 1995. ACM.

\bibitem{upnp1_0}
{UPnP Forum}.
\newblock {UPnP} device architecture 1.0, April 2008.

\bibitem{upnp1_1}
{UPnP Forum}.
\newblock {UPnP} device architecture 1.1, October 2008.

\bibitem{urbani2009reasoning}
Jacopo Urbani, Spyros Kotoulas, Eyal Oren, and Frank van Harmelen.
\newblock Scalable distributed reasoning using {MapReduce}.
\newblock In {\em Proceedings of the ISWC '09}, volume 5823 of {\em Lecture
  Notes in Computer Science}. Springer, 2009.

\end{thebibliography}

\chapter{Background}

\label{ch:background}

\section{Data Propagation}

Data propagation is a concurrent programming paradigm proposed by
Alexey Radul \cite{radul2009thesis,radul2009art}.  Based on the
principles of message-passing concurrency, data propagation relies on
messages sent over networks of monotonic computational blocks, called
``propagators'', and state-bearing ``cells''.  Propagator networks
perform computation when an update of a cell's state triggers the
execution of the propagators that are connected to it.  These
propagators may then update the state of other cells, causing further
propagation.

Although the state of a cell may change, its value may only be
assigned once.  This may seem to be a contradiction at first glance,
but the value assigned to a cell may be a partial value, rather than
one that has been fully-specified.  This allows for continued
modification of its state.  When we discuss ``changing'' a cell's
state, or ``updating'' its contents, what we really mean is that the
partial value of a cell has been updated with additional data.

Cells are initially created with an ``empty'' partial value.  This
special partial value represents a complete absence of knowledge of a
cell's contents and provides no constraints on the value of the cell.
A cell with an empty partial value closely resembles an unassigned
variable, except that it can be easily identified as such during
execution.

% TODO: Depict examples visually?

Partial values stored in a cell need not be objects with partially
defined fields, such as a \texttt{cons} pair in Lisp with a
\texttt{cdr} that has been marked for delayed evaluation
\cite{friedman1976lazy}.  Partial values in a propagator network may,
in fact, be any value, including sets and ranges that may be refined
through appropriate mathematical operations.

The process of refining the partial value of a cell is dependent on a
user-defined merge operation.  This merge operation unifies an
existing partial value with additional information provided in an
``update'' message sent to the cell.  While the merge operation may be
simple assignment when a cell is empty, merge operations are more
interesting when they operate on cells that are not empty.

\begin{figure}
\begin{center}
\includegraphics{cell-merge.pdf}
\caption{A cell applying a merge operation to merge two temperature
  readings.}
\label{fig:cell-merge}
\end{center}
\end{figure}

For example, a cell might contain a range of temperature values that
represents the current air temperature recorded by a thermometer,
including its measurement error.  If a second thermometer is connected
to the cell and sends its \textit{own} measurement as an update, the
merge operation of the cell may take the intersection of the two
temperature ranges to obtain a more precise estimate of the current
air temperature than either thermometer gives on its own.  The cell
may then take the intersection as its new partial value, as depicted
in Figure~\ref{fig:cell-merge}.

A merge operation may be arbitrarily complex.  If a cell is capable of
performing logical deductions to construct its complete contents, we
might be able to employ a merge operation that removes redundant facts
for more efficient storage.  For example, such a cell might contain
two facts: ``Socrates is a man'' and ``Socrates is mortal''.  If the
cell were to receive an update stating that ``All men are mortal'', it
could employ a logically-consistent merge operation that would not
only add the fact ``All men are mortal'' to the cell, but also remove
``Socrates is mortal'', as it may be concluded from the other two
facts.  Despite the fact that the facts stored in the cell are not
strictly increasing (i.e. the cell is not strictly monotonic), if the
cell can still conclude ``Socrates is mortal'', the actual knowledge
that the cell contains is still monotonically increasing.

\begin{figure}
\begin{center}
\includegraphics[width=15cm]{data-propagation.pdf}
\caption{Data propagation at work: a temperature converter.}
\label{fig:data-propagation}
\end{center}
\end{figure}

% TODO: Should I mention stateless vs. monotonic?

Once a cell has merged an update and its partial value has changed,
propagators connected to the cell wake up and perform computation.
These propagators will use the newly updated partial value where they
originally used the cell's previous contents.  During computation,
these propagators may generate updates for \textit{additional} cells.
For example, in Figure~\ref{fig:data-propagation} above, an update to
the top cell at time (b) causes the \texttt{temp-converter} propagator
to fire at time (c).  Upon receiving these updates, these cells will
apply their own merge operations to update their partial data,
starting another merge-and-compute cycle.  This can be seen above when
the propagator at time (c) causes an update to the bottom cell at time
(d).  These repeated cycles of merging and propagation of data drive
computation within a propagator network.

No constraints are placed on the structure of propagator networks.
For example, cycles may be used in propagator networks to create
standard looping control-flow structures.  By allowing a propagator to
update a cell that indirectly contributed to its being fired, the
propagator may ensure that it fires again to include additional
information.  Other propagators may prevent the forwarding of updates
when the input cells meet some criteria.  Such propagators would halt
propagator loops, allowing for switch-like control-flow structures and
finite loops.

We may also perform recursive computation with propagator networks.
As propagators are opaque computational units, and networks of
propagators also perform computation, we may define a class of
propagators called ``compound propagators''.  These compound
propagators may construct a propagator network when they are first
triggered.  This new propagator network may then act as a part of the
original network, and may perform computation based on the original
input cells of the compound propagator by mapping those input cells
appropriately to the newly constructed propagator.

Constructing such networks on demand permits recursive computation; a
compound propagator need only include itself as one of the propagators
in its constructed sub-network to do so.  Propagation of data may
cause as many recursive expansions of the propagator as necessary
simply by repeatedly waking each recursive copy of the compound
propagator in turn.

Ordering of propagation events is undefined, outside of the implicit
cause-effect chain caused by the propagate-and-merge cycle.
Furthermore, propagator networks feature a strong modularity of
computation that is enforced by strictly separating cell state from
propagator computations.  Together, these properties make data
propagation a reasonable candidate for parallel processing and
distributed computation.

% TODO: I almost feel like there should be more citations here.

\section{Representational State Transfer}

\label{sec:rest}

Representational State Transfer, or REST, is an architectural style
for network-based software architectures based on the principles of
hypertext and the HyperText Transfer Protocol (HTTP) and developed by
Roy Fielding \cite{fielding2000thesis}.  Architectures that make use
of Representational State Transfer, also called ``RESTful
architectures'', are designed to store and maintain opaque data
objects, called ``resources'', that represent an abstract concept or
another, more concrete, data object.  Resources generally have four
operations that may be performed on them: creation, retrieval, update,
and deletion.  Together, these four operations are known as CRUD
operations.

Since an abstract concept associated with a resource may not be easily
serialized, RESTful architectures are designed to be platform- and
content-agnostic through the use of resource ``representations''.  For
example, a data set of hourly temperatures may have an image
representation in the form of a graph in GIF format, as well as a
tabular representation in plain-text.  The four fundamental CRUD
operations may be performed in a RESTful architecture by using any
number of these resource representations to specify the relevant
aspects of that resource.

More complex operations on resources, such as those often implemented
as remote procedure calls, may be modelled in a RESTful architecture
by applying the fundamental resource operations with appropriate
representations.  By restricting the number of fundamental operations
and allowing the use of different resource representations, a RESTful
architecture reduces the complexity of software implementations that
use it.  Software uses of a RESTful architecture need only be able to
perform the basic operations with representations they are prepared to
handle; there is no need to support all representations of a resource,
only those relevant to the application.

In the doctoral thesis that first laid out the principles of REST,
Fielding defines an architectural style as ``a coordinated set of
architectural constraints that restricts the roles/features of
architectural elements and the allowed relationships among those
elements within any architecture that conforms to this style.''  In
accordance with this principle, a proper RESTful architecture adheres
to five constraints:

\begin{enumerate}
\item REST assumes a \textit{client-server model} of communication.
  This separates the resources on the server from the software that
  modifies them.  It also determines how data may be stored; only the
  server must find a suitable intermediate format for storage.
  Clients may use any representations they find appropriate.
\item RESTful designs are \textit{stateless}.  Information transmitted
  between the client and server of a RESTful architecture should be
  sufficient to perform the desired operation without maintaining any
  additional state (such as session variables) on the server or
  client.
\item Representations of resources transmitted as part of an operation
  should be \textit{cache-able}.  This constraint is particularly
  useful in terms of the retrieval operation.  By caching information
  that has been previously retrieved, future requests may be optimized
  by referring to a nearby cache, rather than waiting on the central
  server.  It also reduces the burden on the server, as many requests
  may be serviced by a cache instead.
% THOUGHT: This does, however, reduce the usefulness of caches when
% there are a lot of updates.  Expand on this.
\item RESTful designs should employ a \textit{uniform interface}.  By
  standardizing the interface by which resource operations are
  performed, many different client and server implementations may be
  deployed in an inter-operable way.  A uniform interface also reduces
  client and server complexity.
\item Finally, RESTful architectures are \textit{layered
  architectures}.  This allows for the separation of network
  administration from the application itself, providing increased
  scalability.  Caches, back-ends, and proxies may be implemented on
  the network layer without restricting the functionality of the
  application itself.
\end{enumerate}

\begin{table}
\begin{center}
\begin{tabular}[c]{llp{6.5cm}}
CRUD Operation & HTTP Method & RESTful Meaning \\
\hline
Creation & PUT & Create a new resource based on the representation presented. \\
Retrieval & GET & Fetch the desired representation of an existing resource. \\
Update & POST & Modify an existing resource based on a given (partial) representation. \\
Deletion & DELETE & Delete an existing resource. \\
\hline
\end{tabular}
\caption{The four CRUD operations and corresponding HTTP methods.}
\label{tab:http-methods}
\end{center}
\end{table}

Most REST-based applications use the Hypertext Transfer Protocol
(HTTP) as the underlying network protocol of their RESTful
architecture.  HTTP handles data transfers as a series of requests for
data and responses to those requests.  Each request may be made for a
resource identified by its Uniform Resource Locator (URL).  These
requests are also associated with the HTTP method corresponding to the
operation desired.  Four HTTP methods are usually used to implement
REST design: PUT, GET, POST, and DELETE.  These methods are aligned
with the CRUD operations of creation, retrieval, updating, and
deletion, as outlined in Table~\ref{tab:http-methods}.  The PATCH
method \cite{rfc5789} may also be used in place of the POST method to
perform updates.

I have chosen to implement a propagator network by employing RESTful
techniques for several reasons:

\begin{enumerate}
\item Propagator networks need minimal modification to adhere to the
  five constraints of REST.  The stateless constraint in REST is
  stricter than the computational monotonicity of propagators in
  propagator networks.  This means that it is possible to model
  RESTful communication as a propagator in a propagator network.
  Furthermore, the implicit atomicity of update messages sent to cells
  makes these messages well-suited to being representations of
  resources in our implementation.
\item If we assume that update messages are the representations being
  transferred, it seems possible to consider cells to be the resources
  within the design.  Just as cells are the only state-bearing objects
  within propagator networks, resources are the only state-bearing
  objects in a RESTful architecture.
\item Enforcing uniformity of interface in an implementation of a
  distributed propagator network creates an extensible system.  My
  implementation of distributed propagation, DProp, is built using
  Python and has been tested for use on GNU/Linux systems.  By requiring
  that DProp use standard HTTP for cell synchronization, it is
  possible to construct a heterogeneous propagator network that has
  nodes that run various different operating systems and propagator
  implementations.  This makes DProp a suitable platform for
  distributed systems, which already assume a heterogeneous network.
\end{enumerate}

% TODO: Provided that all nodes adhere to the protocol.

Together, these points make Representational State Transfer a
desirable architectural design on which to build our system for
distributed propagation.
